\chapter{Complete Call Lists}
\label{app:lists}

\section{Mainstream and Plus Calls}

\begin{supertabular}{ll}
\call{acey deucey} & \ref{sec:aceydeucey} \nextline
\call{bend the line (from 2-faced line)} & \x \nextline
\call{bend the line (from lines back-to-back)} & \ref{sec:bendthelinebb} \nextline
\call{box the gnat} & \o \nextline
\call{california twirl} & \o \nextline
\call{cast off 3/4} & \x \nextline
\call{chain down the line (from 2-faced line)} & \o \nextline
\call{chain down the line (from LH waves)}} & \x \nextline
\call{chase right} & \x \nextline
\call{circle to a line} & \ref{sec:circletoaline} \nextline
\call{circulate (all 8)} & \ref{sec:all8circ} \nextline
\call{circulate (box)} & \x \nextline
\call{circulate (column)} & \ref{sec:columncirculate} \nextline
\call{cloverleaf} & \ref{sec:cloverleaf} \nextline
\call{coordinate} & \ref{sec:coordinate} \nextline
\call{couples hinge} & \x \nextline
\call{couples trade} & \o \nextline
\call{cross fold} & ? \nextline
\call{cross run} & \x \nextline
\call{crossfire} & \o \nextline
\call{cut the diamond} & \o \ref{sec:diamonds} \nextline
\call{diamond circulate} & \x \ref{sec:diamonds} \nextline
\call{dive thru} & \ref{sec:divethru} \nextline
\call{dixie style to a wave} & \x \nextline
\call{double pass thru} & \ref{sec:doublepassthru} \nextline
\call{eight chain thru} & \ref{sec:tradeby} \nextline
\call{explode and\ldots} & \o \nextline
\call{explode the wave} & \o \nextline
\call{extend} & \ref{sec:extend} \nextline
\call{fan the top (from wave)} & \x \nextline
\call{fan the top (from 2-faced line)} & \o \nextline
\call{ferris wheel} & \ref{sec:ferriswheel} \nextline
\call{flip the diamond} & \o \ref{sec:diamonds} \nextline
\call{(reverse) flutter wheel} & \x \nextline
\call{fold} & ? \nextline
\call{follow your neighbor} & \x \nextline
\call{grand swing thru} & ? \nextline
\call{half sashay} & \o \nextline
\call{half tag} & \ref{sec:tagtheline} \nextline
\call{hinge (from a miniwave)} & \x \nextline
\call{hinge (from a couple)} & \o \nextline
\call{2 ladies chain} & \x \nextline
\call{lead right/left} & \x \nextline
\call{linear cycle (from wave)} & \o \nextline
\call{load the boat} & \o \nextline
\call{partner tag (from a miniwave)} & \x \nextline
\call{partner tag (from a couple)} & \o \nextline
\call{pass the ocean} & \o \nextline
\call{pass thru} & \o \nextline
\call{pass to the center} & \ref{sec:passtothecenter} \nextline
\call{peel off} & ? \nextline
\call{peel the top} & \o \nextline
\call{ping pong circulate} & ? \nextline
\call{put centers in} & ? \nextline
\call{right and left thru} & \o \nextline
\call{recycle (from waves)} & \x \nextline
\call{relay the deucey} & \o \nextline
\call{roll} & ? \nextline
\call{roll away} & \o \nextline
\call{centers/ends (of a wave or line) run} & \x \nextline
\call{scoot back} & \o \nextline
\call{separate} & ? \nextline
\call{single circle to a wave} & \o \nextline
\call{slide thru} (from boys facing girls) & \o \nextline
\call{spin chain and exchange the gears} & ? \nextline
\call{spin chain the gears} & ? \nextline
\call{spin chain thru} & ? \nextline
\call{spin the top} & \x \nextline
\call{split the outsides} & ? \nextline
\call{spread} & \o \nextline
\call{square thru $\langle n\rangle$} & \o \nextline
\call{star thru} & \o \nextline
\call{step thru} & \o \nextline
\call{step to a wave} & \o \nextline
\call{sweep 1/4} & \x \nextline
\call{swing thru} & \x \nextline
\call{tag the line} & \ref{sec:tagtheline} \nextline
\call{touch 1/4} & \x \nextline
\call{trade} & \o \nextline
\call{track 2} & \ref{sec:track2} \nextline
\call{centers (of a wave or line) trade} & \x \nextline
\call{trade by} & \ref{sec:tradeby} \nextline
\call{trade the wave} & \o \nextline
\call{triple scoot} & \o \nextline
\call{turn thru} & \o \nextline
\call{u-turn back (everyone)} & \o \nextline
\call{veer left/right} & \o \nextline
\call{walk and dodge} & \o \nextline
\call{wheel and deal (from 2-faced line)} & \o \nextline
\call{wheel and deal (from lines back-to-back)} & \ref{sec:wheelanddealbb} \nextline
\call{wheel around} & \o \nextline
\call{zoom} & ?
\end{supertabular}

\section{Advanced Calls}

\begin{supertabular}{ll}
\call{box counter rotate} & \x \nextline
\call{box transfer} & \x \nextline
\call{cross-over circulate} & \ref{sec:xovercirc} \nextline
\call{half breed thru} & \o \nextline
\call{lockit} (from wave) & \o \nextline
\call{lockit} (from 2-faced line) & \x \nextline
\call{mix} & \o \nextline
\call{pair off} & \o \nextline
\call{pass the sea} & \o \nextline
\call{quarter in/out} & \o \nextline
\call{quarter right/left} & \x \nextline
\call{quarter thru} & \o \nextline
\call{recycle} (from facing couples) & \x \nextline
\call{scoot and dodge} & \x \nextline
\call{scoot and weave} & \o \nextline
\call{slide} & \o \nextline
\call{slip} & \x \nextline
\call{slither} & \o \nextline
\call{square chain thru} & \x \nextline
\call{swap around} & \x \nextline
\call{swing} & \o \nextline
\call{switch the wave} & \o \nextline
\call{switch to a diamond} & \x \nextline
\call{three quarter thru} & \o \nextline
\call{trade circulate} & \ref{sec:tradecirc} \nextline
\call{turn and deal} (from 2-faced line) & \o \nextline
\call{wheel thru} & \x
\end{supertabular}

\section{C-1 Calls}

\begin{supertabular}{ll}
\call{flip the line} & \nextline
\call{2/3 recycle} &\nextline
\call{flip back} &\nextline
\call{flip your neighbor} &\nextline
\call{tag back to a wave} &\nextline
\call{tag your neighbor} &
\end{supertabular}

